Dear Editor,

Thank you for considering our manuscript MS 4760992181296929 “In vitro anticoagulant and antioxidant activities of Jatropha gossypiifolia L. (Euphorbiaceae) leaves aiming therapeutical applications” by Juliana Félix-Silva, Thiago Souza, Rafael B. G. Camara, Bárbara Cabral, Arnóbio A. Silva-Júnior, Ivanise M. M. Rebecchi, Silvana M. Zucolotto, Hugo A. O. Rocha and myself, for publication in BMC Complementary and Alternative Medicine Journal.

We have carefully read the editorial point addressed by the Editor and have answered it in this letter and performed the required changes in the revised manuscript.

We hope this revised manuscript is now suitable for publication in BMC Complementary and Alternative Medicine Journal.

Yours sincerely,

Matheus F. Fernandes-Pedrosa, Ph.D
Department of Pharmacy
Federal University of Rio Grande do Norte
Phone: +55.84.3342-9820 Fax: +55.84.33429822
E-mail: mpedrosa@ufrnet.br

Editorial comments:

Please confirm whether consent for the taking and use of human blood in your
study was obtained from the donors, and include a statement to this effect in the Methods section of your manuscript.

Our answer:

The blood was collected from the human donors after a written informed consent has been obtained, according to the research protocol approved by our institution (protocol no.092/09). This statement was included in Methods section of the manuscript, as requested.